Author’s response to reviews

Title: Socio-demographic determinants and effect of structured personal diabetes care: a 19-year follow-up of the randomized controlled study Diabetes Care in General Practice (DCGP)

Authors:
Andreas Heltberg (anhe@sund.ku.dk)
Volkert Siersma (siersma@sund.ku.dk)
John Andersen (sal@sund.ku.dk)
Christina Ellervik (ellervik@gmail.com)
Henrik Brønnum-Hansen (henrik.bronnum-hansen@sund.ku.dk)
Jakob Kragstrup (jakr@sund.ku.dk)
Niels de Fine Olivarius (olivarius@sund.ku.dk)

Version: 3 Date: 25 Nov 2017

Author’s response to reviews:

Dear Stavros Liatis

BMC Endocrine Disorders

ATT: BEND-D-17-00167

Dear Editor-in-chief and Reviewers

We thank the editor-in-chief, associate editor, and reviewers for the valuable and constructive comments.

On behalf of the authors I am sending this revised manuscript, which we hope that you will agree for publication in The BMC Endocrine Disorders.

We have made revisions accordingly; including format, declarations, author contributions, contact information, description of additional material and other revisions mentioned.
Best regards

Andreas Heltberg
MD, GP and PhD-student

Department of General Practice
Institute of Public Health
Faculty of Health Sciences
University of Copenhagen
Øster Farimagsgade 5 Q
1014 Copenhagen
Denmark

E-mail: anhe@sund.ku.dk, tel +4524412624